Author’s response to reviews

Title: Construction and validation of a dimensional scale exploring mood disorders: MAThyS (Multidimensional Assessment of Thymic States)

Authors:

Chantal Henry (chantal.henry@inserm.fr)
Katia Mbailara (katia_mb@hotmail.com)
Flavie Mathieu (flavie.mathieu@inserm.fr)
Rollon Poinson (rpoinsot@institutcancer.fr)
Bruno Falissard (falissard_b@wanadoo.fr)

Version: 4 Date: 2 July 2008

Author’s response to reviews:

Here is the version with formatting changes requested